Author's response to reviews

Title: A novel BRCA-1 mutation in Arab kindred from east Jerusalem with breast and ovarian cancer

Authors:

Luna Kadouri (luna@hadassah.org.il)
Dani Bercovich (danib@migal.org.il)
Arava Elimelech (aravaeli@gmail.com)
Israela Lerer (lereris@hadassah.org.il)
Michal Sagi (msagi@hadassah.org.il)
Gila Glusman (gglusman@hadassah.org.il)
Chen Shochat (chens@migal.org.il)
Sigal Korem (sigalk@migal.org.il)
Tamar Hamburger (tamarh@hadassah.org.il)
Aviram Nissan (anissan@hadassah.org.il)
Nahil Abu-Halaf (abухhalafs@yahoo.com)
Muhmud Badryyah (mbadriya@yahoo.com)
Dvorah Abeliovich (dvoraha@cc.huji.ac.il)
Tamar Peretz (tamarp@cc.huji.ac.il)

Version: 5  Date: 18 January 2007

Author’s response to reviews:

18.1.2007

To the Editor,

We corrected and adapted the text to the BMC cancer format as listed:

1. websites were referenced in text and table legened.

2. references were put in squar brackets

3. we corrected to 'Authors’ contribution'

Regards,

Luna Kadouri